Author's response to reviews

Title: Accuracy of self-reported height measurements in parents and its effect on mid-parental target height calculation

Authors:

Ieva Braziuniene (Braziuniene-Leva@cooperhealth.edu)
Thomas A Wilson (tawilson@mail.som.sunysb.edu)
Andrew H Lane (andrew.lane@stonybrook.edu)

Version: 3 Date: 23 March 2007

Author's response to reviews: see over
Andrew H. Lane, M.D.
Associate Professor of Pediatrics
Division of Pediatric Diabetes and Endocrinology
Department of Pediatrics, T-11, 080
Health Sciences Center
SUNY Stony Brook
Stony Brook, NY 11794-8111
(631) 444-3429

3/22/07

Dear Sirs,

Thank you for accepting in principle our manuscript # 1893570862125025, “Accuracy of self-reported height measurements in parents and its effect on mid-parental target height calculation” for BioMed Central. I have just resubmitted the manuscript with the requested minor revisions.

Please contact me with any further questions.

Sincerely,

Andrew H. Lane, M.D.